Author's response to reviews

Title: Obstructive jaundice caused by a biliary mucinous cystadenoma in a young female: a case report

Authors:

Pramodh C Chandrasinghe (pramodh@slt.net.lk)
Chandika H Liyanage (chandikaliyanage@hotmail.com)
Kemal I Deen (radihan@mail.ewisl.net)
Ruwan E Wijesuriya (ruwanwije@hotmail.com)

Version: 5 Date: 4 September 2013

Author's response to reviews: see over
Dear Editor,

Re-submission: Obstructive jaundice caused by a biliary mucinous cystadenoma in a young female: a case report

We wish to resubmit the corrected version of the above mentioned case report. The corrections are in different colours and underlined.

The following corrections were made.

Reefree 1:

1. Added the details of the histology report of the cyst including a description of its lining epithelium and characteristic subepithelial stroma in the case report.
2. In the discussion section a paragraph was added on differentiating the tumour from other cysts in the liver histologically plus newer reports on immunohistochemistry methods.

Refree 2:

1. Rephrasing of several sentences as suggested by the referee was done and are in blue and underlined.
2. Normal value of CA 19.9 was added.
3. Follow up details were added to the case report.
4. Several spelling errors as pointed out by the referee were corrected.

Editors suggestions

1. The title was rephrased.
2. J Young et al article on different radiological features was referred to in the article in relation to the case report in the discussion section
3. The age and ethnicity of the patient was mentioned in the abstract.
4. Key words were added
5. Acknowledgements were made.
6. List of abbreviations was added
7. A disclosure on payments received by the authors was added.
8. Dates were removed from the figures
9. Figures were removed from the text and submitted as a separate file.
10. We were not able to find a publishable ultrasound image of the lesion unfortunately.

Hope we have made the necessary amendments to the satisfaction of the referees.

Kind regards

Sincerely

P C Chandrasinghe